Author's response to reviews

Title: Evaluating the use of fibrin glue for sealing of low-output enterocutaneous fistulas: study protocol for a randomized controlled trial

Authors:

Xiuwen Wu (xiuwenwoo@gmail.com)
Jianan Ren (jiananr@gmail.com)
Gefei Wang (wqfwang@gmail.com)
Jianzhong Wang (surgeonwiz@163.com)
Feng Wang (winer_77@163.com)
Yueping Fan (fanyp900164@sina.com)
Yuanxin Li (liyuanxin1966@163.com)
Gang Han (gmail1973@163.com)
Yanbing Zhou (shougencao@gmail.com)
Xiaofei Song (xiaofei81920@163.com)
Bin Quan (surgeonzhao@163.com)
Min Yao (minyao99@hotmail.com)
Jieshou Li (jieshou_li@163.com)

Version: 4 Date: 8 September 2015

Author’s response to reviews: see over
Dear Editor,

We would like to express our warmest thanks to you for all these constructive and positive comments. We appreciate your time and efforts to improve this manuscript very much! We have revised the protocol, and would like to re-submit it for your consideration. All formatting requests have been addressed in the revised manuscript and have been highlighted.

Again, thank you for providing these constructive comments. We hope that this revised version is now acceptable for publication in Trials. Thanks a lot for sacrificing your time in the process of revision again.

Your consideration for this manuscript is highly appreciated.

Look forward to hearing from you.

Sincerely yours,
Jinan Ren, MD, FACS
Department of Surgery, Jinling Hospital
Medical School of Nanjing University
305 East Zhongshan Road, Nanjing, 210002, China.